Author's response to reviews

Title: Prevention of morbidity in sickle cell disease - qualitative outcomes, pain and quality of life in a randomized crossover pilot trial of overnight supplementary oxygen and auto-adjusting continuous positive airways pressure (POMS 2a): study protocol for a randomized controlled trial

Authors:

Jo Howard (Jo.Howard@gstt.nhs.uk)
Baba Inusa (Baba.Inusa@gstt.nhs.uk)
Christina Liossi (C.Liossi@soton.ac.uk)
Eufemia Jacob (eufemia@sonnet.ucla.edu)
Patrick B Murphy (Patrick.b.murphy@kcl.ac.uk)
Nicholas Hart (Nicholas.Hart@gstt.nhs.uk)
Johanna C Gavlak (Johanna.Gavlak@uhs.nhs.uk)
Maria Chorozoglou (M.Chorozoglou@soton.ac.uk)
Sati Sahota (s.sahota@ucl.ac.uk)
Carol Nwosu (Carol.Nwosu@scyss.org)
Maureen Gwam (maureengwam@hotmail.co.uk)
Atul Gupta (atulgupta@nhs.net)
David C Rees (David.Rees@kcl.ac.uk)
Swee Lay Thein (sl.thein@kcl.ac.uk)
Isabel C Reading (I.C.Reading@soton.ac.uk)
Fenella J Kirkham (Fenella.Kirkham@ucl.ac.uk)
Man Ying E Cheng (M.Y.Cheng@soton.ac.uk)

Version: 6 Date: 22 July 2015

Author's response to reviews:

Thank you for further review of the manuscript and for your provisional acceptance

I have made the author contributions clearer by putting names rather than initials. I apologise for the typographical error for Man Ying Edith Cheng's contribution which I have now corrected

I have added a note about our additional logo file after the references

We look forward to hearing from you

Yours sincerely

Fenella Kirkham